Author's response to reviews

Title: Stability of response characteristics of a Delphi panel: application of bootstrap data expansion

Authors:

Ralitsa B. Akins (ralitsa_akins@yahoo.com)
Homer Tolson (htolson@tamu.edu)
Bryan R. Cole (b-cole@tamu.edu)

Version: 3 Date: 31 October 2005

Author's response to reviews:

To:
The BioMed Central Editorial Team
BioMed Central Research Methodology

October 31, 2005

Re: Final Copyedit of MS 9409921797521902 - Stability of Response Characteristics of a Delphi Panel: Application of Bootstrap Data Expansion

Dear BioMed Central Editorial Team:

We would like to thank you for accepting for publication our manuscript titled, "Stability of Response Characteristics of a Delphi Panel: Application of Bootstrap Data Expansion" and for this opportunity to submit the final copyedit of our paper.

Yours truly,

Ralitsa Akins, MD, PhD
Associate Research Scientist
Rural and Community Health Institute
Quality and Patient Safety Initiatives
The Texas A&M University System Health Science Center
ralitsa_akins@yahoo.com